Author's response to reviews

Title: Genetic variability in CYP3A4 and CYP3A5 in primary liver, gastric and colorectal cancer patients

Authors:

Guillermo Gervasini (ggervasi@unex.es)
Elena Garcia-Martin (elenag@unex.es)
Jose M Ladero (jladero.hcsc@salud.madrid.org)
Rosa Pizarro (rosapiarro@unex.es)
Javier Sastre (jsastre.hcsc@salud.madrid.org)
Carmen Martinez (cmartine@unex.es)
Monserrat Garcia (Monse.garcia@unex.es)
Manuel Diaz-Rubio (mdiazrubio.hcsc@salud.madrid.org)
Jose AG Agundez (jagundez@unex.es)

Version: 5 Date: 2 July 2007

Author's response to reviews:

Sirs,

Thank you for your e-mail. I uploaded the revised manuscript with the following changes:
The title was written in sentence case.
The affiliations were written in full.
The keywords were removed.
The red text was written in black font.
The abstract was structured as required.
The section headings were written in sentence case.
The website addresses were cited as required.
The section "Patients, materials and methods" were renamed as required.

We hope that you consider the revised version of the manuscript adequate for publication in BMC Cancer.
We are looking forward to hear from you.

Yours sincerely,

Prof. Jose A.G. Agundez
Department of Pharmacology
Medical School, University of Extremadura
Avda de Elvas s/n. 06071, Badajoz, Spain
Phone/Fax +34924289458, E-mail: jagundez@unex.es